Author’s response to reviews

Title: Serum levels of soluble programmed death-ligand 1 (sPD-L1) in patients with primary central nervous system diffuse large B-cell lymphoma

Authors:

Inju Cho (injucho81@gmail.com)
Hansang Lee (helios0516@hanmail.net)
Sang Yoon (dtd43@hanmail.net)
Kyung Ryu (kjryu1203@gmail.com)
Young Hyeh Ko (yhko310@skku.edu)
Won Kim (wskimsme@skku.edu)
Seok Jin Kim (kstwoh@skku.edu)

Version: 2 Date: 30 Jan 2020

Author’s response to reviews:

Editor Comments:

1. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

-> We added the full name of the ethics committee that approved the study and the committee’s reference number.

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

--> We describe that in the funding section.

Reviewer comments

Herve Tilly, MD (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated
versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.


--> We revised the reference as you recommended.